Author’s response to reviews

Title: Insufficiency of prosthetic posterolateral overlap related to recurrence after laparoscopic transabdominal preperitoneal inguinal hernioplasty, as assessed by video review

Authors:
Masanori Sato (msnrsato@hama-med.ac.jp)
Masashi Nozawa (miyabic_n@yahoo.co.jp)
Takahiro Watanabe (watanabetop78@yahoo.co.jp)
Takanobu Onoda (continue_trying@icloud.com)
Atsuko Matsuyama (a256116447@yahoo.co.jp)
Norihiko Shiiya (shiyanor@hama-med.ac.jp)
Hidetoshi Wada (wadahide@municipal-hospital.shimada.shizuoka.jp)

Version: 2 Date: 24 Jan 2020

Author’s response to reviews:

Dear, Ms Megan Barling,

Thank you for your polite and kind cooperation regarding the manuscript.

Upon your request, we revised the manuscript as follows:

1. Corresponding Author Email

-- Please ensure the corresponding author email address is included on the title page of your manuscript page and ensure this matches the email address on our system.

>> I put the email address right after the address of the affiliation.

2. Table 1 Identifying

-- We note that table 1 contains potentially identifiable information. It is BioMed Central policy to not publish more than 2 indirect identifiers without explicit consent for publication from the participants, as per this paper
(http://trialsjournal.biomedcentral.com/articles/10.1186/1745-6215-11-9). Currently, the age and gender information in conjunction with other details, may compromise patient anonymity.

>> We changed the patient age to age-ranges.

3. Consent

-- In the section 'Ethics Approval and Consent to Participate', please state whether the informed consent obtained was written or verbal. If verbal, please state the reason and whether the ethics committee approved this procedure.

>> We obtained "Opt-out" consent which was approved by the ethic committee, where consents were not always necessary to perform the study and publish the article. We added the "Opt-out" consent was approved by the ethic committee.

4. Availability of Data and Materials

-- The Availability of data and materials section refers to the raw data used in your study and presenting tables and figures is not sufficient to state that all data is contained within the manuscript and additional files.

>> We changed the sentence as follows:

• The datasets used and/or analysed during the current study available from the corresponding author on reasonable request.

5. Authors Contributions

-- Please clarify whether AT should be AM.

>> We revised AT to AM.

6. Figure 1 Attribution

-- Please state in the cover letter whether the image depicted in figure 1 is your own or taken from another source. If taken from another source please acknowledge the source in the figure legend, and if it is under copyright also state the written permission given to use and adapt it.

>> The figure was drew by MS, not taken from another source.

7. Conclusion
-- Please insert a conclusion heading.

>> We changed according to your indication.

8. Formatting

-- Please ensure that all figures/tables and supplementary files are cited within the text. Any items which are not cited may be deleted by our production department upon publication.

>> We checked them all.

-- Please remove the COI files from your additional files.

>> We did.

Again, thank you for your cooperation

Sincerely,

Best

Masanori Sato, MD, PhD